Author’s response to reviews

Title: Long-term efficacy of infliximab for refractory ulcerative colitis: Results from a single center experience

Authors:

Satoshi Yamada (yamadasa@kuhp.kyoto-u.ac.jp)
Takuya Yoshino (tyocci@kuhp.kyoto-u.ac.jp)
Minoru Matsuura (minomats@kuhp.kyoto-u.ac.jp)
Naoki Minami (nminami@kuhp.kyoto-u.ac.jp)
Takahiko Toyonaga (toyo.necco@gmail.com)
Yusuke Honzawa (honzawa@kuhp.kyoto-u.ac.jp)
Yoshihisa Tsuji (ytsuji@kuhp.kyoto-u.ac.jp)
Hiroshi Nakase (hiropy_n@kuhp.kyoto-u.ac.jp)

Version: 5
Date: 13 April 2014

Author’s response to reviews:

Dear Editorial Board

Thank you very much for accept our arranged manuscript entitled, “Long-term efficacy of infliximab for refractory ulcerative colitis: Results from a single center experience”.

According to your suggestion, we addressed our manuscript based on copy editing with English proofreading. The point-by-point responses to the reviewer’s comments on the first revisions are added in the latter of this cover letter. All arrangements according to the first reviewer’s comments were indicated in the text by red colored changes of modified words or sentences.

It would be very much appreciative if you kindly review our addressed manuscript after copy editing, and accept it for publication in BMC gastroenterology.

Sincerely yours

Hiroshi Nakase, MD, PhD, Lecture
Department of Gastroenterology and Hepatology, Graduate school of Medicine, Kyoto University
54 Shogoin, Kawahara-cho, Sakyo-ku, Kyoto, 606-8507, Japan
Tel: +81-75-751-4319
Fax: +81-75-751-4303
E-mail: hiropy_n@kuhp.kyoto-u.ac.jp